\documentclass[10pt]{amsart}
\usepackage[cp1251]{inputenc}
\usepackage[english,russian]{babel}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{graphicx}
\usepackage{cite}

\def\udcs{510.5} %Here the author places classificators of the paper according to Russian classification system
\def\mscs{03D25} %Here the author places  classificators according to the AMS classification list.
\setcounter{page}{144}

\newtheorem{lemma}{Lemma}
\newtheorem{theorem}{Theorem}
\newtheorem{corollary}{Corollary}
\newtheorem{proposition}{Proposition}
\newtheorem*{theorem3}{Theorem 3}
\theoremstyle{definition}
\newtheorem{definition}{Definition}


\newcommand{\nc}[1]{\mathrm{#1}}
\newcommand{\mc}[1]{\mathcal{#1}}
\newcommand{\mb}[1]{\mathbb{#1}}
\newcommand{\mf}[1]{\mathfrak{#1}}
\newcommand{\N}{\mathbb N}
\newcommand{\dg}[1]{\mathbf{#1}}
\newcommand{\ang}[1]{\big\langle #1\big\rangle}
\newcommand{\nl}{\nolimits}
\newcommand{\TrU}[2]{\triangle_{#1}(#2)}
\newcommand{\dfn}{\stackrel{\textrm{\clap{\scriptsize def}}}{=}}

\def\logo{{\bf\huge S\raisebox{0.2ex}{\hspace{0.55ex}\raisebox{0.05ex}e\hspace{-1.65ex}$\bigcirc$}MR}}

\def\semrtop
     {
  \vbox{
     \noindent\logo
     \hspace{80mm}\raisebox{1ex}{ISSN 1813-3304 }

     \vspace{5mm}

     \begin{center}
     {\huge СИБИРСКИЕ \ ЭЛЕКТРОННЫЕ} \\[2mm]  %Headings in Russian
     {\huge МАТЕМАТИЧЕСКИЕ ИЗВЕСТИЯ} \\[2mm]
     {\large Siberian Electronic Mathematical Reports} \\[1mm]
     {\LARGE\tt{http://semr.math.nsc.ru}}\\[0.5mm]
%     {\small Институт математики им. С.Л.Соболева СО РАН}\\[-1mm]
%     {\small Sobolev Institute of Mathematics SB RAS}
     \end{center}
     \vspace{-3mm}
     \noindent
     \begin{tabular}{c}
     \hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa} \\
     \hline\hline
     \end{tabular}

     \vspace{1mm}
     {\flushleft\it Том 16, стр. 144--144 (2019) \hspace{65mm}{\rm\small УДК \udcs}} %Volume, pages in Russian,
                                                                                    %filled by Editorial board
     \newline
         {\rm\small DOI~10.33048/semi.2019.16.xxx}\hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa}{\rm\small MSC\ \ \mscs }
  }%\vbox
}

\begin{document}
\begin{thebibliography}{1}

\bibitem{N09} 
A. Nies,
{\it Computability and randomness}, Oxford University Press, New York, 2009.

\bibitem{N06}
A. Nies, 
{\it Reals which compute little}. In Z. Chatzidakis, P. Koepke, $\&$ W. Pohlers (Eds.), Lo\-gic Colloquium '02 (Lecture Notes in Logic, pp. 261--275). Cambridge, Cambridge University Press, 2006.

\bibitem{DGW07}
N. Greenberg, R. Downey, R. Weber,
{\it Totally $\omega$-computably enumerable degrees and bounding critical triples},
Journal of Mathematical Logic, {\bf 7} (2007), 145--171.

\bibitem{S63}
G.E. Sacks,
 {\it On the degrees less than $\dg 0'$}, Annals of Mathematics, {\bf 77} (1963), 211--231.
 
 \bibitem{S87}
R.I. Soare, 
  {\it Recursively enumerable sets and degrees.} A study of computable functions and computably generated sets. Perspectives in mathematical logic. 
  Springer-Verlag, Berlin, Heidelberg, New York, etc., 1987.

\bibitem{W80}
L. Welch, 
{\it A hierarchy of families of recursively enumerable degrees and a theorem of founding minimal pairs},
 Ph.D. Thesis, University of Illinois, Urbana, 1980.
 
\bibitem{BF82}
M. Bickford, F. Mills,
{\it Lowness properties of r.e. sets}, Manuscript, UW Madison, 1982.
 
\bibitem{D09} 
 D. Diamondstone, 
 {\it Promptness does not imply superlow cuppability},
 Journal of Symbolic Logic, {\bf 74} (2009), 1264--1272.
 
 \bibitem{F11}
 M. Faizrakhmanov, 
 {\it A semilattice generated by superlow computably enumerable degrees}, 
 Russian Mathematics, {\bf 55} (2011), 74--78.
 
\bibitem{DN18} 
 R. Downey, K.M. Ng,
{\it Splitting into degrees with low computational strength},
Annals of Pure and Applied Logic, {\bf 169} (2018), 803--834.

\bibitem{M86} 
 J. Mohrherr, 
 {\it A refinement of low $n$ and high $n$ for the r.e. degrees}, Z. Math. Logik
Grundlag. Math., {\bf 32} (1986), 5--12.

\bibitem{A11}
M.M. Arslanov, {\it The Ershov hierarchy}. In S. B. Cooper $\&$ Andrea Sorbi (eds.), Computability in Context: Computation and Logic in the Real World. World Scientific, 2011.
  
\end{thebibliography}

\renewcommand{\refname}{References}
\renewcommand{\proofname}{Proof.}
\renewcommand{\figurename}{Fig.}


\thispagestyle{empty}

\title[Splitting of c.e. degrees and superlowness]{Splitting of c.e. degrees and superlowness}

\author{{M.Kh. Faizrahmanov}}%
\address{Marat Khaidarovich Faizrahmanov
\newline\hphantom{iii} Kazan (Volga Region) Federal University,
\newline\hphantom{iii} Volga Region Scientific-Educational Centre of Mathematics,
\newline\hphantom{iii} str. Kremlevskaya, 35,
\newline\hphantom{iii} 420008, Kazan, Russia}%
\email{marat.faizrahmanov@gmail.com}


\thanks{\sc Faizrahmanov, M.Kh., 
One elementary difference between the classes of low and superlow c.e. degrees}
\thanks{\copyright \ 2015 Faizrahmanov M.Kh.}
\thanks{\rm The work is supported by the Russian Science Foundation (grant 22-21-20024, https://rscf.ru/project/22-21-20024/) and performed under the development program of Volga Region Mathematical Center (agreement 075-02-2022-882).}
\thanks{\it Received  January, 1, 2015, published  March, 1,  2015.}%


\semrtop \vspace{1cm}
\maketitle {\small
\begin{quote}
\noindent{\sc Abstract. } 
In this paper, we show that for any superlow c.e. degrees $\dg a$ and $\dg b$ there exists a superlow c.e. degree $\dg c$ such that $\dg c\not=\dg a_0\cup\dg b_0$ for
all c.e. degrees $\dg a_0\leqslant\dg a$, $\dg b_0\leqslant\dg b$. This provides one more elementary difference between the classes of low c.e. degrees and superlow c.e. degrees.
We also prove that there is a c.e. degree that is not the supremum of any two superlow not necessarily c.e. degrees degrees.
\medskip

\noindent{\bf Keywords:} low degree, superlow degree, jump-traceable set.
 \end{quote}
}


\section{Introduction}

In the last fifty or so years, there have emerged a large number of lowness
notions associated with c.e. (as well as other classes of) sets and degrees.
Some notable examples include the superlow
degrees, the jump-traceable degrees, the array computable degrees,  the $K$-trivial degrees, the contiguous degrees, the strongly jump-traceable degrees (for all of these definitions, see~\cite{N09}).
Some of these concepts will, where
necessary, be defined later in context.


In this paper, we study
and compare two lowness properties, being low and being superlow.
The first problem to which the paper is devoted is the search for new elementary differences between the classes of low c.e. degrees and superlow c.e. degrees. 
This problem was formulated by Nies~\cite{N06} and solved by Downey, Greenberg and Weber~\cite{DGW07}. They proved that no superlow c.e. degree can bound a critical triple, but some low c.e. degree can.
To find another elementary difference, we will use the following theorem, which is derived from the Sacks Splitting Theorem~\cite{S63}\cite[Ch. VII, Theorem 3.2]{S87}:
\begin{theorem}[Welch~\cite{W80}]\label{WelchTh}
There exist low c.e. degrees $\dg a$ and $\dg b$ such that for every c.e. degree $\dg c$ there are c.e. degrees $\dg a_0\leqslant\dg a$ and $\dg b_0\leqslant\dg b$
such that $\dg c=\dg a_0\cup\dg b_0$.
\end{theorem}
\noindent The proof of Theorem~\ref{WelchTh} is based on the fact that, by the Sacks Splitting Theorem, $\varnothing'$ is a disjunctive union of two non-intersecting low c.e. sets (hence, $\dg 0'=\dg a\cup\dg b$ for some low c.e. degrees $\dg a$ and $\dg b$). 
Although Bickford and Mills~\cite{BF82}\cite[Theorem~6.1.4]{N09} have shown that $\dg 0'=\dg a\cup\dg b$ for some superlow c.e. degrees $\dg a$ and $\dg b$, in Theorem~\ref{WelchTh}, 
lowness cannot be replaced by superlowness.
This follows from the first result of this paper:
\begin{theorem3}
Let $\dg a$ and $\dg b$ be superlow c.e. degrees. Then there is a superlow c.e. degree $\dg c$ such that $\dg c\not=\dg a_0\cup\dg b_0$ for
all c.e. degrees $\dg a_0\leqslant\dg a$, $\dg b_0\leqslant\dg b$.
\end{theorem3}
\noindent Thus, Theorems~\ref{WelchTh} and 3 immediately provide another elementary difference between the classes of low c.e. degrees and superlow c.e. degrees.

There are other elementary properties that distinguish the classes of
low and superlow c.e. degrees. For instance, Dia\-mond\-stone~\cite{D09} proved that there exists a noncappable (hence, low
cuppable) degree which does not cup with a superlow c.e. degree to $\dg 0'$. In~\cite{F11}, it was proved that the semilattice generated by superlow c.e. degrees
and the semilattice of all c.e. degrees (that is generated by low c.e. degrees) are not elementary equivalent. This implies that there is a low c.e degree which is not the supremum
of any two superlow c.e. degrees. Downey and Ng~\cite{DN18} proved that such a degree can be chosen to be ultrahigh. In this paper, we prove that there is a c.e. degree that is not the supremum of any two superlow
not necessarily c.e. degrees. The question of the existence of such degrees was also formulated in~\cite{N06}.

Recall that a set $A$ is \textit{low} if its jump $A'$
is Turing-below the halting problem $\varnothing'$. The following concept is more restrictive.
\begin{definition}
The set $A$ is \textit{superlow} if $A'\leqslant_{tt}\varnothing'$.
Equivalently, $A'(e) =\lim_sg(e, s)$ for a computable $0, 1$-valued $g$ such that $g(e, s)$ changes at most $b(e)$
times, for a computable function $b$.
\end{definition}
\noindent This notion goes back to work of Mohrherr~\cite{M86}, and an unpublished manuscript of
Bickford and Mills~\cite{BF82} (where only superlow c.e. sets are studied, which are called there <<abject>>).

We write $J^A(e)$ for $\Phi_{e}(A;e)$, the jump at argument $e$.
While lowness and su\-per\-low\-ness restrict the \textit{domain} $A' = \{e : J^A(e)\downarrow\}$ of $J^A$,
\textit{jump traceability} introduced by Nies~\cite{N06,N09} expresses that $J^A(e)$ has few possible values.
\begin{definition}
A c.e. sequence of sets $\{T_e\}_{e\in\N}$ is a \textit{trace} if for some computable $h$,
$$\forall e\,[|T_e|\leqslant h(e)].$$
 The set $A$ is \textit{jump-traceable} if there is a trace $\{T_e\}_{e\in\N}$ such that
$$\forall e\,[J^A(e)\downarrow\Rightarrow J^A(e) \in T_e].$$
\end{definition}
\noindent In the proof of the first result of this paper, we essentially use the following theorem.
\begin{theorem}[Nies~\cite{N06}]
 Superlowness and jump-traceability coincide within the c.e. sets.
\end{theorem}
\noindent Note that none of the properties implies the other within the $\omega$-c.e. sets.

Our notation from computability theory is mostly standard. In the following, $\varphi^{(n)}_e$ denotes the $n$-ary partial computable function with the G\"{o}del number $e$.
For unary partial computable functions we skip the upper index.
We write $\varphi^{(n)}_e(x_1,\dots,x_n)\downarrow$ if this computation is defined, and $\varphi^{(n)}_e(x_1,\dots,x_n)\uparrow$ otherwise.
We are assuming that  
$$\varphi^{(n)}_{k,s}(x_1,\dots,x_k)\downarrow\Rightarrow\varphi^{(n)}_k(x_1,\dots,x_k)<s.$$ 
Let $\Phi_e$ be the Turing functional with the G\"{o}del number $e$ and $u(X;e,x,s)$ be the $use$-function of the computation $\Phi_{e,s}(X;x)$.
We are assuming that 
$$\Phi_{e,s}(X;x)\downarrow\Rightarrow\max\{\Phi_{e,s}(X;x),u(X;e,x,s)\}<s.$$
Let $W_e(X)$ be the domain of $\Phi_e(X)$.
 For unexplained notions we refer to Soare~\cite{S87}

\section{The elementary difference}

In the proof of the following theorem, we let $c(x, y)$ denote the computable
pairing function $2^x(2y + 1) - 1$. We write $c(x_0,\dots, x_{n+2})$ instead of\linebreak $c(x_0, \dots, x_n,c(x_{n+1},x_{n+2}))$.
Let $\N^{[y]}=\{c(x,y):x\in\N\}$ for each $y$.

\begin{theorem}\label{split}
Let $\dg a$ and $\dg b$ be superlow c.e. degrees. Then there is a superlow c.e. degree $\dg c$ such that $\dg c\not=\dg a_0\cup\dg b_0$ for
all c.e. degrees $\dg a_0\leqslant\dg a$, $\dg b_0\leqslant\dg b$.
\end{theorem}
\begin{proof}
Let $\dg a=\deg_T(A)$ and $\dg b= \deg_T(B)$. We will construct a superlow c.e. set $C$ satisfying the following requirements:
$$N_e:\exists^{\infty} s\,[\Phi_{e,s}(C_s;e)\downarrow]\Rightarrow\Phi_{e}(C;e)\downarrow,$$
$$R^{e,i,j}_{k,p,q}:W_p=\Phi_{i}(A)\,\&\,W_q=\Phi_{j}(B)\,\&\,W_p\oplus W_q=\Phi_k(C)\Rightarrow C\not=\Phi_{e}(W_p\oplus W_q).$$
To satisfy the  requirements $N_e$, we will use the standard lowness strategy from the low simple set construction~\cite[Ch. VII, Theorem 1.1]{S87}.

The strategy for satisfying the requirement $R^{e,i,j}_{k,p,q}$ is as follows. First we note that all c.e. sets $A_0$, $B_0$ with $A_0\leqslant_TA$, $B_0\leqslant_TB$
 are uniformly (by Turing functionals providing these reducibilities) superlow. 
 Therefore, one can choose integers $z$ and $m$ such that if the left-hand side of the implication in $R^{e,i,j}_{k,p,q}$ is true, then the sum of the number of changes in the values 
 $J^{W_p}(z)$ and $J^{W_q}(z)$ of the jump operator does not exceed $m$.
We then choose witnesses 
$$x_0<x_1<\dots<x_m$$ 
 to diagonalize the equality
 $$C=\Phi_{e}(W_p\oplus W_q)$$
  in such a way that adding each $x_i$, $0<i\leqslant m$, to $C$ does not affect the restraints imposed by $x_{i-1}$.
Thus, by sequentially adding the witnesses $x_m,x_{m-1},\dots$ to $C$ and changing the values of $J ^{W_p}(z)$ and $J^{W_q}(z)$, we get 
$$C\not=\Phi_{e}(W_p\oplus W_q).$$
 In more detail,
to satisfy the requirement $R^{e,i,j}_{k,p,q}$, we construct auxiliary functionals $\Theta$ and $\Xi$ with use-functions $\theta$ and $\xi$ respectively.
Using the Recursion theorem, we can assume that before starting the construction we know indices $a$ and $b$ such that $\Theta=\Phi_a$ and $\Xi=\Phi_b$. Hence, we know c.e. sequences of sets 
$\{T^{i,p}_x\}_{i,p,x\in\N}$, $\{U^{j,q}_x\}_{j,q,x\in\N}$
and computable sequences of functions $\{f^i_p\}_{i,p\in\N}$, $\{g^j_q\}_{j,q\in\N}$ such that 
\begin{equation}\label{t1}
|T^{i,p}_x|\leqslant f^i_p(x),\ |U^{j,q}_x|\leqslant g^j_q(x),
\end{equation}
\begin{equation}\label{t2}
W_p=\Phi_{i}(A)\,\&\,\Theta(W_p;x)\downarrow\Rightarrow\Theta(W_p;x)\in T^{i,p}_x,
\end{equation}
\begin{equation}\label{t3}
W_q=\Phi_{j}(B)\,\&\,\Xi(W_q;x)\downarrow\Rightarrow\Xi(W_q;x)\in U^{j,q}_x
\end{equation}
for all $i,j,p,q,x$.
Now we choose a $z$ from some preselected computable set and define $m=f^i_p(z)+g^j_q(z)+1$.
Then we proceed according to the following instructions.

\medskip
\noindent 1. Choose a finite sequence $x_0,\dots,x_m$ such that
$$\Phi_{e,s}(W_{p,s}\oplus W_{q,s};x_n)\downarrow=0\ \mbox{for all}\ n\leqslant m,$$
$$\Phi_{k,s}(C_s;x)\downarrow\ \mbox{for each}\ x\leqslant\max\{u(W_{p,s}\oplus W_{q,s};e,x_n,s):n\leqslant m\},$$
$$x_{n+1}>\max\{u(C_s;k,x,s):x\leqslant u(W_{p,s}\oplus W_{q,s};e,x_n,s)\}\ \mbox{for each}\ n<m,$$
 at some stage $s$.
We define 
$$\Theta(W_{p,t};z)\downarrow=\Xi(W_{q,t};z)\downarrow=s$$ at each stage $t\geqslant s$ and the values of their 
$use$-functions at $z$ equal to $s$.
 
\medskip
\noindent 2. 
We do not allow elements less than or equal to $s$ to be enumerated into $C$.
 
 \medskip
\noindent 3. We start sequentially enumerating the elements $x_m, x_{m-1}, \dots$ into $C$ whenever  
$$\Theta(W_{p,t};z)\in T^{i,p}_{z,t},\ \Xi(W_{q,t};z)\in U^{j,q}_{z,t}$$
for some new stage $t$.

 \medskip
\noindent 4. We define
$$\Theta(W_{p,u+1};z)=\Theta(W_{p,t};z)+1,\ \Xi(W_{q,u+1};z)=\Xi(W_{q,t};z)+1$$
for suitable $t$ and $u$
whenever $x_n$ is enumerates into $C$.

\medskip\noindent Taking into account conditions (\ref{t1})--(\ref{t3}) and the equality $m=f^i_p(z)+g^j_q(z)+1$, we will have that the requirement $R^{e,i,j}_{k,p,q}$ is satisfied.

\bigskip
\noindent\textit{Construction}

\medskip
\noindent Stage 0. Let $\gamma_0=\varnothing$, $C_0=\varnothing$, $h_0(y)=r(y,0)=0$ for each $y$. At each next stage, we assume that the value of each parameter remains the same, unless otherwise specified.

\medskip
\noindent Stage $2s+1=2c(e,i,j,k,p,q,v)+1$. Let $y=c(e,i,j,k,p,q)$, $z=c(y,h_s(y))$, and $m=f^i_p(z)+g^j_q(z)+1$.

\medskip
\noindent 1. If $\gamma_s(z)\uparrow$ and there is a finite sequence $x_0,\dots,x_m\in\N^{[y]}\setminus C_s$ such that
$$\Phi_{e,s}(W_{p,s}\oplus W_{q,s};x_n)\downarrow=0\ \mbox{and}\ x_n>r(w,s)\ \mbox{for all}\ n\leqslant m,\ w\leqslant 2y,$$
$$\Phi_{k,s}(C_s;x)\downarrow\ \mbox{for each}\ x\leqslant\max\{u(W_{p,s}\oplus W_{q,s};e,x_n,s):n\leqslant m\},$$
$$x_{n+1}>\max\{u(C_s;k,x,s):x\leqslant u(W_{p,s}\oplus W_{q,s};e,x_n,s)\}\ \mbox{for each}\ n<m,$$
then we define 
$$\gamma_{s+1}(z)=c(m,x_0,\dots,x_m),$$ 
$$r(2y+1,s+1)=s,$$
$$\Theta(W_{p,s};z)=\Xi(W_{q,s};z)=s$$
with the use-values
$$\theta(W_{p,s};z)=\xi(W_{q,s};z)=s$$
respectively,
and go to the next stage. We are assuming that 
$$\theta(W_{p,t};z)=\xi(W_{q,t};z)=s$$
for each $t\geqslant s$.

\medskip
\noindent Suppose $\gamma_s(z)\downarrow=c(m,x_0,\dots,x_m)$.

\medskip
\noindent 2. If  there are $n\leqslant m$, $w\leqslant 2y$ such that $x_n\leqslant r(w,s)$,
then we define $$h_{s+1}(z)=h_{s}(z)+1$$ and go to the next stage.

\medskip
\noindent 3. Suppose
$$\Theta(W_{p,s};z)\downarrow\in T^{i,p}_{z,s},\ \Xi(W_{q,s};z)\downarrow\in U^{j,q}_{z,s},$$
$$\forall n\leqslant m\,[\Phi_{e,s}(W_{p,s}\oplus W_{q,s};x_n)=C_s(x_n)],$$
and there is an $l\leqslant m$ such that $x_l\not\in C_s$.
Let $l_0=\max\{l\leqslant m:x_l\notin C_s\}$.
 If 
  $$\Phi_{k,s}(C_s;x)=(W_{p,s}\oplus W_{q,s})(x)$$ for each
 $x\leqslant\max\{u(W_{p,s}\oplus W_{q,s};e,x_{l_0},s):n\leqslant l_0\}$,
 then we define $$C_{s+1}=C_s\cup\{x_{l_0}\},\ r(2y+1,s+1)=s,$$ and go to the next stage.

\medskip
\noindent 4. Assume that 
$$\Theta(W_{p,t};z)\downarrow,\ \Xi(W_{q,t};z)\downarrow,$$
where $t=c(e,i,j,k,p,q,v-1)$.
 Then we define 
 $$\Theta(W_{p,s};z)=\Theta(W_{p,t};z)+1,\ \mbox{if}\ W_{p,s}\upharpoonright\theta(W_{p,t};z)\not=W_{p,t}\upharpoonright\theta(W_{p,t};z),$$
  $$\Xi(W_{q,s+1};z)=\Xi(W_{q,t};z)+1,\ \mbox{if}\ W_{q,s+1}\upharpoonright\xi(W_{q,t};z)\not=W_{q,t}\upharpoonright\xi(W_{q,t};z),$$
 and go to the next stage. 


\medskip
\noindent Stage $2s+2$. Define
$$r(2e,s+1)=u(C_s;e,e,s)$$
for each $e$.

\medskip
\noindent\textit{End construction}

\bigskip
It remains to check that $C$ is superlow. Indeed, we know the number of wit\-nes\-ses~$z$ belonging to a preselected computable set needed to satisfy each $R$-requirement. 
Therefore, we know the number of restraints imposed by each requirements of lower priority. Hence $C$ is superlow. Note that similar arguments were used in the construction of a superlow simple set~\cite[Theorem 1.6.5]{N09}.
 \end{proof}
 \begin{corollary}
 The classes of low c.e. and superlow c.e. degrees are not elementary equivalent.
 \end{corollary}
 \noindent The following corollary shows that in the Sacks Splitting Theorem~\cite{S63} \cite[Ch. VII, Theorem 3.1]{S87}, in the general case, splitting sets cannot be chosen to be superlow.

 \begin{corollary}
 There are no superlow c.e. sets $A$ and $B$ such that $A\cap B=\varnothing$ and $A\cup B=\varnothing'$.
 \end{corollary}
 \begin{proof} 
 Assume that $\varnothing'=A\cup B$ for some superlow c.e. sets  $A$ and $B$ with $A\cap B=\varnothing$. Then there exist superlow c.e. sets $\widehat{A}$ and $\widehat{B}$ such that
 $$K_0=\{c(e,x):x\in W_e,\ e\in\N\}=\widehat{A}\cup\widehat{B},\  \widehat{A}\cap\widehat{B}=\varnothing.$$
 Let $n$ be an integer such that $W_n\not\equiv_TA_0\oplus B_0$ for all c.e. sets $A_0\leqslant_T\widehat{A}$ and $B_0\leqslant_T\widehat{B}$.
 We have
 $$W_n\equiv_T\{c(n,x):c(n,x)\in\widehat{A}\}\oplus\{c(n,x):c(n,x)\in\widehat{B}\}.$$
 That is a contradiction.
 \end{proof}



\section{A c.e. degree that cannot be split into two superlow degrees}

For a binary function $\varphi^{(2)}_e$, we let
$$\widehat{\varphi}_e(x)=
\begin{cases}
\varphi^{(2)}_e(x,\min\{n:\varphi^{(2)}_e(x,n)\downarrow\}), & \mbox{if}\ \exists n\,[\varphi^{(2)}_e(x,n)\downarrow],\cr
\mbox{undefined}, & \mbox{otherwise}.
\end{cases}
$$
It follows from~\cite{A11} that $X$ is superlow iff there exists an $e$ such that $X'=\widehat{\varphi}_e$ iff for every $i$ there exists an $e$ such that $W_i(X)=\widehat{\varphi}_e$.
\begin{theorem}
There exists a c.e. degree $\dg c$ such that $\dg c\not=\dg a\cup\dg b$ for all superlow degrees $\dg a$ and $\dg b$.
\end{theorem}
\begin{proof}
We will construct a c.e. set $C$ and Turing functionals $\Theta$, $\Xi$ satisfying the following requirements:
$$R^{e,i,j}_{k,p,q}: C=\Phi_e(\Phi_i(C)\oplus\Phi_j(C))\,\&\,\Phi_i(C)\oplus\Phi_j(C)=\Phi_k(C)\Rightarrow$$
$$\Rightarrow H(\Phi_i(C))\not=\widehat{\varphi}_p\,\vee\,S(\Phi_j(C))\not=\widehat{\varphi}_q,$$
where $H(X)=\nc{dom}\,\Theta(X)$ and $S(X)=\nc{dom}\,\Xi(X)$.

The strategy for satisfying the requirement $R^{e,i,j}_{k,p,q}$ is similar to the strategy from Theorem~\ref{split}.
The main difference between these strategies is as follows. We need to diagonalize the condition 
$$H(\Phi_i(C))=\widehat{\varphi}_p\,\&\,S(\Phi_j(C))=\widehat{\varphi}_q,$$
but we cannot immediately choose $z$ and $m$ as in Theorem~\ref{split} (that is why $C$ cannot be made superlow). Instead, we are looking for $z$, $m_p$, $m_q$ such that $\varphi_{p,s}(z,m_p)\downarrow$ and $\varphi_{q,s}(z,m_q)\downarrow$.
After that we choose witnesses
$$x_0<x_1<\dots<x_m,$$
where  $m=m_p+m_q+1$,
for the diagonalization
in such a way that adding each~$x_i$, $0<i\leqslant m$, to $C$ does not affect the restraints imposed by $x_{i-1}$.
Thus, by sequentially adding the witnesses $x_m,x_{m-1},\dots$ to $C$ and changing the values of $H(\Phi_i(C))$ and $H(\Phi_j(C))$, we get 
$$H(\Phi_i(C))\not=\widehat{\varphi}_p\,\vee\,S(\Phi_j(C))\not=\widehat{\varphi}_q.$$
In more detail, to satisfy the requirement $R^{e,i,j}_{k,p,q}$, we proceed according to the following inst\-ruc\-ti\-ons.

\medskip
\noindent 1. Choose integers $z$ and $m_p,m_q$ such that $\varphi_{p,s}(z,m_p)\downarrow$ and $\varphi_{q,s}(z,m_q)\downarrow$. Let $m=m_p+m_q+1$.

\medskip
\noindent 2. Choose a finite sequence $x_0,\dots,x_m$ such that
$$\Phi_{e,s}(\Phi_{i,s}(C_s)\oplus\Phi_{j,s}(C_s);x_n)=0,$$
$$\Phi_{k,s}(C_s;x)\downarrow\ \mbox{for each}\ x\leqslant\max\{u(\Phi_{i,s}(C_s)\oplus\Phi_{j,s}(C_s);e,x_n,s):n\leqslant m\},$$
$$x_{n+1}>\max\{u(C_s;k,x,s):x\leqslant u(\Phi_{i,s}(C_s)\oplus\Phi_{j,s}(C_s);e,x_n,s)\}\ \mbox{for each}\ n<m.$$

\medskip
\noindent 3. 
We do not allow elements less than or equal to $s$ to be enumerated into $C$.


\medskip
\noindent 4. We start sequentially enumerating the elements $x_m, x_{m-1}, \dots$ into $C$ whenever
$$\widehat{\varphi}_{p,s}(z)=H_s(\Phi_{i,s}(C_s))(z)=\widehat{\varphi}^{(2)}_{q,s}(z)=S_s(\Phi_{j,s}(C_s))(z)=1.$$

\medskip\noindent Taking into account the equality $m=m_p+m_q+1$, we will have that the requirement $R^{e,i,j}_{k,p,q}$ is satisfied.

\bigskip
\noindent\textit{Construction}

\medskip
\noindent Stage 0. Let $\gamma_0=\varnothing$, $C_0=\varnothing$, $h_0(y)=r(y,0)=0$ for each $y$. At each next stage, we assume that the value of each parameter remains the same, unless otherwise specified.

\medskip
\noindent Stage $s+1=c(e,i,j,k,p,q,v)+1$. Let $y=c(e,i,j,k,p,q)$, $z=c(y,h_s(y))$. Let
$$
m=
\begin{cases}
m_p+m_q+1, & \mbox{if}\ m_p=\min\{n:\varphi_{p,s}(z,n)\downarrow\},\ m_q=\min\{n:\varphi_{q,s}(z,n)\downarrow\},\cr
-1, & \mbox{if}\ \forall n\,[\varphi_{p,s}(z,n)\uparrow]\,\vee\,\forall n\,[\varphi_{q,s}(z,n)\uparrow].
\end{cases}
$$.

\medskip
\noindent 1. If $\gamma_s(z)\uparrow$ and there is a finite sequence $x_0,\dots,x_m$ such that
$$\Phi_{e,s}(\Phi_{i,s}(C_s)\oplus\Phi_{j,s}(C_s);x_n)=0\ \mbox{and}\ x_n>\max\{r(w,s):w<y\}\ \mbox{for each}\ n\leqslant m,$$
where $\max\varnothing=-1$,
$$\Phi_{k,s}(C_s;x)\downarrow\ \mbox{for each}\ x\leqslant\max\{u(\Phi_{i,s}(C_s)\oplus\Phi_{j,s}(C_s);e,x_n,s):n\leqslant m\},$$
$$x_{n+1}>\max\{u(C_s;k,x,s):x\leqslant u(\Phi_{i,s}(C_s)\oplus\Phi_{j,s}(C_s);e,x_n,s)\}\ \mbox{for each}\ n<m,$$
then we define 
$$\gamma_{s+1}(z)=c(m,x_0,\dots,x_m),$$ 
$$r(y,s+1)=s,$$
$$\Theta(\Phi_{i,s}(C_s);z)=\Xi(\Phi_{j,s}(C_s);z)=0$$
with the use-values
$$\theta(\Phi_{i,s}(C_s);z)=\xi(\Phi_{j,s}(C_s);z)=s$$
respectively,
and go to the next stage. We are assuming that 
$$\theta(\Phi_{i,t}(C_t);z)=\xi(\Phi_{j,t}(C_t);z)=s$$
 for each $t\geqslant s$.

\medskip
\noindent Suppose $\gamma_s(z)\downarrow=c(m,x_0,\dots,x_m)$.

\medskip
\noindent 2. If  there are $n\leqslant m$, $w<y$ such that $x_n\leqslant r(w,s)$,
then we define $$h_{s+1}(z)=h_{s}(z)+1$$ and go to the next stage.

\medskip
\noindent 3. Assume that
$$\varphi^{(2)}_{p,s}(z,m_p)=H_s(\Phi_{i,s}(C_s))(z),\  \varphi^{(2)}_{q,s}(z,m_q)=S_s(\Phi_{j,s}(C_s))(z).$$
If $$z\not\in H_s(\Phi_{i,s}(C_s))\ \mbox{or}\ z\not\in S_s(\Phi_{j,s}(C_s)),$$
then we define
$$\Theta(\Phi_{i,s}(C_s);z)=0\ \mbox{or}\ \Xi(\Phi_{j,s}(C_s);z)=0$$
respectively. If
$$z\in H_s(\Phi_{i,s}(C_s))\cap S_s(\Phi_{j,s}(C_s))$$
and there exists an $l\leqslant m$ such that $x_l\not\in C_s$, then we define
$$C_{s+1}=C_{s}\cup\{x_{l_0}\}$$
for the greatest $l_0$ with $x_{l_0}\not\in C$, and
$$r(y,s+1)=s.$$

\medskip
\noindent \textit{End construction}

\medskip
\noindent It is not hard to see that the construction combines strategies of satisfying each $R$-requirement by the finite injury priority method.
This completes the proof of the theorem.
\end{proof}

Note that we can easily combine the satisfying of the $R$-requirements from last theorem with the satisfying of the $N$-requirements form Theorem~\ref{split}.
Hence, there exists a low c.e. degree $\dg c$ such that $\dg c\not=\dg a\cup\dg b$ for all superlow degrees $\dg a$ and $\dg b$.

\bigskip

\end{document}
